Author's response to reviews

Title: Relationship between estimated glomerular filtration rate, albuminuria, and oxidant status in the Japanese population

Authors:

Yuko Ishizaka (yishizaka-circ@umin.ac.jp)
Minoru Yamakado (amakado-m@mitsuihosp.or.jp)
Akiko Toda (0ry76651385074n@ezweb.ne.jp)
Mizuki Tani (tani-m@mitsuihosp.or.jp)
Nobukazu Ishizaka (nobuishizka-tky@umin.ac.jp)

Version: 4 Date: 28 July 2013

Author's response to reviews:

July 29, 2013

Dear Editors:

We are sending a third manuscript entitled "Relationship between estimated glomerular filtration rate, albuminuria, and oxidant status in the Japanese population" by Yuko Ishizaka, Minoru Yamakado, Akiko Toda, Mizuki Tani, and myself for the publication in BMC Nephrology as Original Article.

As the comment from the reviewers pointed out no particular sentences and context, what we have made in this latest update of the cited references.

Thank you in advance for your consideration. I am looking forward to hearing from you in the near future.

Sincerely yours,

Nobukazu Ishizaka, MD

Department of Cardiology, Osaka Medical College,
Takatsuki-shi Daigaku-machi 2-7, Osaka 569-8686, Japan

e-mail: ishizaka@poh.osaka-med.ac.jp
Telephone: +81-72-683-1221 (Japan)
Facsimile: +81-72-684-6533 (Japan)